Author's response to reviews

Title: Frequency of self-medicated antibiotics, factors leading to their administration and knowledge of their adverse effects among non-medical university students of Karachi: A cross-sectional study

Authors:

Syed Jawad Shah (ssjawadss@gmail.com)
Hamna Ahmad (hamnaahmad278@gmail.com)
Rija Binte Rehan (rija.rehan@live.com)
Sidra Najeepb (sdrnajeeb@gmail.com)
Mirrah Mumtaz (mirrahmumtaz@yahoo.com)
Muhammad Hashim Jillani (hashimjilani@gmail.com)
Muhammad Sharoz Rabbani (sharoz.rabbani88@gmail.com)
Muhammad Zakariya Alam (z_alam_8@hotmail.com)
Saba Faroq Awan (sabafaroog92@gmail.com)
Muhammad Masood Kadir (masood.kadir@aku.edu)

Version: 2
Date: 23 July 2014

Author's response to reviews:

Dear Editor-in-Chief,

I am sending you the revised version of our manuscript entitled “Frequency of self-medicated antibiotics, factors leading to their administration and knowledge of their adverse effects among non-medical university students of Karachi: A cross-sectional study”.

We made a few changes to the document. First, we changed the name of one of our authors as she asked us that her official name was not that one which we submitted prior to it. We also made separate parts on the manuscript for conclusion, acknowledgement and competing interests. We took the tables out and put them at the end and also made a separate portion just before the references for figure titles and legends although we just have one figure. For the figure, we made a separate document and uploaded it.

These are the changes that we made so far as suggested by you.

Thank You.

Syed Jawad Shah
Final Year Medical Student,
Aga Khan University Medical College, Karachi, Pakistan